

JAMES Journal of Advances in Modeling Earth Systems

RESEARCH ARTICLE

10.1029/2018MS001540

This article is a companion to Haynes et al. (2019) http://doi.org/10.1029/ 2018MS001541.

Key Points:

- Grassland phenology can be predicted using a strategy based on growth stages that respond to climate and weather
- Dynamic prognostic phenology links biophysical and biogeochemical processes to predict the terrestrial carbon cycle

Supporting Information:

Supporting Information S1

Correspondence to:

K. D. Haynes, Katherine.Haynes@colostate.edu

Citation:

Haynes, K., Baker, I. T., Denning, S., Stöckli, R., Schaefer, K., Lokupitiya, E. Y., & Haynes, J. M. (2019). Representing grasslands using dynamic prognostic phenology based on biological growth stages: 1. Implementation in the Simple Biosphere Model (SiB4). Journal of Advances in Modeling Earth Systems, 11, 4423–4439. https://doi.org/10.1029/ 2018MS001540

Received 23 OCT 2018 Accepted 18 NOV 2019 Accepted article online 5 DEC 2019 Published online 18 DEC 2019

©2019. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Representing Grasslands Using Dynamic Prognostic Phenology Based on Biological Growth Stages: 1. Implementation in the Simple Biosphere Model (SiB4)

K. D. Haynes¹, I. T. Baker¹, A. S. Denning¹, R. Stöckli², K. Schaefer³, E. Y. Lokupitiya⁴, and J. M. Haynes⁵

¹Department of Atmospheric Science, Colorado State University, Fort Collins, CO, USA, ²Climate Division, Federal Office of Meteorology and Climatology, MeteoSwiss, Zurich, Switzerland, ³National Snow and Ice Data Center, University of Colorado Boulder, Boulder, CO, USA, ⁴Department of Zoology and Environment Sciences, University of Colombo, Sri Lanka, ⁵Cooperative Institute for Research in the Atmosphere, Colorado State University, Fort Collins, CO, USA

Abstract Grasslands grow in a sequence of seasonal growth stages that respond to both climate and weather, and these relationships can be used to establish a strategy for predicting plant phenology. Current plant states (phenophase) can be represented as one of established growth stages that dictate carbon allocation and leaf photosynthetic capacity. Calculating daily phenophases from climate and environmental relationships allows for sequential growth stages (i.e., well-defined seasonal cycles with a single growth period) or dynamic growth stages (i.e., multiple growth periods during a growing season). Senescence results from biomass mortality in response to environmental conditions. This approach uses a single mechanistic framework to represent grassland ecology, removing the dependence on satellite-based vegetation indices and individual site tuning of parameters. Rather than being specified, a variety of properties emerge, from allometric relationships such as root-shoot ratios, to behavior across moisture gradients, to interannual variability in growing season lengths, carbon stores, and land surface fluxes. Using dynamic phenology stages to link biophysical and biogeochemical processes provides a mechanism to predict self-consistent land-atmosphere exchanges of carbon, water, energy, radiation, and momentum, as well as carbon storage in cascading pools of biomass; and describing these processes in a mathematically determinate model makes them clear, testable, and usable for predictions. This paper describes this new phenology method as it is implemented in the Simple Biosphere Model Version 4 (SiB4), and a companion paper evaluates this method at grassland sites worldwide.

Plain Language Summary Every year, grasslands grow in a sequence of seasonal growth stages that have evolved in nature. For grasslands, climate and weather patterns can be used to establish a strategy for predicting these stages. Using five different possible stages, the current stage can be used to determine which part of the plant grows the fastest. This approach uses a single set of equations to represent grassland ecology, removing the dependence on satellite-based vegetation information. Rather than being specified, a variety of properties emerge, such as different growth patterns in regions that receive more rainfall and different growth rates per year. This approach links plant processes and provides a way to model plant growth and its interaction with the atmosphere. This paper describes this new method as it is implemented in the Simple Biosphere Model Version 4 and provides an example at a grassland site.

1. Introduction

Phenology in terrestrial ecosystems, which refers to periodic vegetation events that are driven by environmental conditions, plays an essential role in biogeochemical processes, from determining local land surface properties to impacting regional land-atmosphere interactions, driving the terrestrial carbon cycle, and influencing global climate change (Dahlin et al., 2015; Friedl et al., 2014; Kim et al., 2015; Schwartz, 2013). Plant phenology, which incorporates visible seasonal events while integrating underlying biological functionality, can be easily seen and understood, making it a powerful indicator of climate change; however, since phenology provides a holistic view of biological mechanisms, it is also a powerful tool for scientists. Since there is a critical link between phenology and terrestrial biochemical, biophysical, and spectral properties, studying phenology not only leads to further understanding of ecosystem processes but also can be used to monitor and understand biospheric responses to global climate change (Fu et al., 2014; Gu et al., 2003; Richardson et al., 2013).

Vegetation phenology is a critical component of terrestrial biosphere models (TBMs), which are essential tools for a variety of applications, from predicting land-atmosphere exchanges to improving our understanding of terrestrial ecosystems and predicting impacts of climate change. Although it is well known that phenology is specific to individual species, TBMs provide an opportunity to explore ecosystem-level seasonal patterns and capture broad vegetational responses (Gu et al., 2003). Accurately simulating phenology is essential in TBMs because seasonality controls numerous aspects of plant growth and development, which then regulates ecosystem processes and land-atmosphere fluxes, directly impacting terrestrial biogeochemical and hydrological cycles (Eamus et al., 2016; Friedl et al., 2014; Pau et al., 2011).

To simulate plant phenology, TBMs employ two different methods: using satellite observations directly or predicting phenology using process-based models. While prescribing the phenology from satellite observations takes advantage of global remotely sensed data, these models are limited to simulating time periods when satellite observations are available (Medvigy et al., 2009; Sellers, Randall, et al., 1996). In contrast, TBMs that model phenology have the advantage of predicting the vegetation state at the expense of introducing empirical or calibrated parameters (Chuine et al., 2013; Kim et al., 2015; Lawrence et al., 2011).

Due to the strong applicability to climate change research, interest in including process-based phenology models in TBMs has been expanding. One common approach to predict phenology used by TBMs is a rule-based strategy to predict phenological transitions. Environmental conditions are prominent causes of plant phenology, and of these temperature is strongly related to leaf-out and growing season onset (Schwartz, 2013). In using temperature to time phenological transitions, one technique is to use a growing-degree day approach (Richardson et al., 2012; White et al., 1997). For simulating crops, the growing-degree day determines the developmental stage, which is used to adjust carbon allocation and vegetation characteristics as the growth progresses throughout the season (e.g., Lokupitiya et al., 2009). For simulating natural vegetation, global TBMs use cumulative growing-degree days to identify leaf onset (Lawrence et al., 2011; Krinner et al., 2005). To improve phenological predictions, rule-based models employ chilling requirements for leaf onset (e.g., Kaduk & Los, 2011; Picard et al., 2005), add day of year as an additional parameter (e.g., Fisher et al., 2007), or incorporate soil water and photoperiod for the timing of senescence (e.g., Borchert et al., 2005; Dahlin et al., 2015, 2017; Lawrence et al., 2011).

An alternative to predicting phenological transitions is to continuously characterize leaf phenology. To do this, Jolly et al. (2005) introduced a combined growing season index that utilizes thresholds of temperature, light, and humidity to predict the vegetation state. Stöckli et al. (2008, 2011) used this approach combined with data assimilation to fit parameters at every grid cell and predict present-day phenology. Forkel et al. (2014) adopted this approach in a dynamic vegetation model, replacing the humidity requirement with water availability and adding a heat stress limiting function. In a different approach that still continuously characterizes leaf phenology, Caldararu et al. (2014) developed a mechanistic model of leaf phenology using the hypothesis that phenology is a strategy for optimal carbon gain at the canopy level, making leaf adjustments in response to light, temperature, and soil moisture.

Although work is ongoing to improve phenology predictions, TBMs have systematic errors in the simulation of phenology (Migliavacca et al., 2012). A model intercomparison study evaluated the representation of phenology and the associated seasonality of ecosystem-scale CO_2 exchange (Richardson et al., 2012), finding that existing models are unable to accurately simulate the timing of both the beginning and ending of the growing season and highlighting the need for improved understanding and representation of environmental controls on vegetation phenology. Additionally, a study by Keenan et al. (2012) found that models underrepresent the variability in spring leaf-out, leading to systematic errors in carbon fluxes. When used in a global climate model, TBMs with prognostic vegetation phenology add the capability to represent interannual variability; however, mean biogeophysical simulations tend to be degraded due to the added degree of freedom (Lawrence et al., 2011, 2012).

One of the difficulties in predicting phenology in TBMs comes from the fact that the ecological processes underlying observed phenological patterns at ecosystem scales is still uncertain (Chuine et al., 2013; Liang & He, 2017; Melaas et al., 2016; Pau et al., 2011; Richardson et al., 2013). Satellite remote sensing provides broad spatial and temporal coverage of terrestrial vegetation; however, the attribution of observed land surface phenology to biophysical processes remains lacking (Liang & He, 2017; Richardson et al., 2017). Increasing observational data and real-time monitoring combined with cross-scale data integration and modeling provides an opportunity for improving the understanding of the relationships between phenology patterns and ecological processes from local to regional to global scales (Richardson et al., 2017; Morellato et al., 2016).

We developed a prognostic phenology strategy that uses a combination of predicting rule-based transitions while continuously calculating leaf phenology and the vegetation state in an innovative hybrid approach. This method continuously predicts the current phenological stage (phenophase) by creating a phenology index (*PI*) that is determined from three elements: (1) a growth index (PS_{GI}) that estimates the benefit of growing new leaves; (2) a weather potential (PS_{WX}) that calculates vegetation stress based on current temperature and water availability; and (3) a day length index (PS_{DL}) to obtain developmental responses to photoperiod. The phenophase is used to determine the carbon allocation and maximum photosynthesis rate. Sencescence is simulated from both continual turnover (with rates that respond to assimilation rate and temperature) and increasing transfer fractions (with fractions that depend on daylength, temperature, and water availability). The combination of these elements results in the simulation of key phenological events (such as leaf-out and senescence) while introducing an innovative, dynamic approach. By unifying phenology, carbon pools, and land-atmosphere exchanges, the Simple Biosphere Model (SiB4) predicts self-consistent land surface properties, vegetation characteristics; soil moisture content; carbon storage in cascading pools of biomass; and land-atmosphere exchanges of carbon, water, energy, radiation, and momentum.

This paper describes the new phenology strategy and its implementation in SiB4 (Baker et al., 2008; Sellers et al., 1986; Sellers, Los, et al., 1996; Sellers, Randall, et al., 1996). In representing grasslands, SiB4 uses three plant functional types (PFTs): arctic C3 (C3A), non-arctic C3 (C3G), and C4 (C4G). All parameters used in this methodology are PFT-specific. Section 2 describes the use of phenological stages to predict leaf-out and vegetation states throughout the growing season, and section 3 describes the use of turnover and litterfall processes to predict senescence. Since grazing is important in grassland ecosystems, section 4 briefly describes the implementation of grazing, and section 5 describes how these processes are combined to fully predict the terrestrial carbon cycle. While a thorough evaluation of this approach is in a companion paper, section 6 provides an example of the terrestrial carbon cycle at a temperate grassland site. Finally, section 7 discusses the utility of this approach and provides suggestions for future applications. Supporting information provides justification for the use of 10-day running means (section S1), grazing equations (section S2), and a step-by-step description and illustration of the calculation sequence in SiB4 (section S3), along with tables of the new variables, parameter values, and figures demonstrating the relationships used in the phenology model.

Since this paper only shows results at one site merely to illustrate the methodology, a companion paper (Haynes et al., 2019) thoroughly investigates the terrestrial carbon cycle of 55 grassland sites worldwide, using what we believe to be an unprecidented collection of grassland data. The companion paper uses remotely sensed leaf area index (LAI), eddy covariance carbon and energy fluxes at towers, and annual net primary productivity at long-term *in situ* sites to evaluate diurnal, seasonal, annual, and interannual variability predicted by SiB4 compared to these data sets while simultaneously investigating large-scale grassland relationships spanning climate zones, including the impact of precipitation gradients and drought on phenology, aboveground vegetation, and productivity. Haynes et al. (2019) shows that SiB4 predicts LAI, carbon, and energy fluxes with root-mean-square errors <15% and individual biases <10% at all available sites. SiB4 is able to predict observed large-scale grassland behavior across precipitation gradients and in response to drought with nearly identical relationships to those seen in the observations.

2. Dynamic Prognostic Phenology

Plant development responds to seasonal and interannual climatic variations (Jarvis, 1976; Larcher, 2002; Stöckli et al., 2008). Seasonal development occurs via physiological processes that cause visual changes in biomass. All plants pass through the main phenological events in an annual cycle, and these visual characteristics can be categorized into phenological stages. While phenology progressions are rhythmical, they are never identical nor exactly periodic; instead, the rates at which phenological changes occur depend on



environmental conditions and differ not only between vegetation types but also in time (Schwartz, 2013). Phenological events are results of internal plant processes, thus continuously predicting the phenophase provides a mechanism to procure not only the visible changes in vegetation, but also the observed changes in physiology and biogeochemistry.

In order to respond to climate and weather conditions, plant growth must depend on environmentally responsive mechanisms from a combination of contributions, including air temperature, precipitation, humidity, radiation, and soil conditions (Chen, 2003). Phenophases can be predicted from climatological and environmental relationships. For natural vegetation, determining the phenophase is challenging as there are numerous confounding issues, including regions with no clear growing season and plants that have multiple growing seasons (Reed et al., 2003).

2.1. Phenological Stages

To predict grassland phenology, SiB4 uses dynamic phenophases that respond to the leaf state and environmental conditions. The overarching idea is that vegetation progresses through phenology stages during a growing season; however, the phenophase must allow plants to progress through the stages at widely different rates or to return to growth stages after midseason stress. For example, desert environments with short rainy seasons rapidly progress through the stages, while tropical grasslands are never dormant and rarely experience full leaf-out since they maintain leaves year-round.

To represent grasslands as they develop and mature through the growing season, this dynamic phenology scheme defines five seasonal stages (nstage = 5):

- 1. *Leaf-Out:* Leaf growth begins from stored carbon, since the canopy cannot support photosynthesis. This stage begins when the environmental conditions become suitable for photosynthesis, which is determined from daylength, soil moisture, and temperature.
- 2. *Growth:* Leaf growth is promoted from carbon allocation, since the canopy benefits by intercepting more light. This stage begins when the canopy is large enough to support photosynthesis.
- 3. *Maturity:* Leaf amount is sustained and allocation is balanced between canopy maintenance, root growth, and product production (seeds, fruit, and flowers). This stage begins when the plant no longer benefits from growing new leaves.
- 4. *Stress*: All carbon is allocated to the roots for storage and leaf photosynthetic capacity (V_{Max}) is reduced. This stage begins when there is a penalty for growth (negative cost for new leaves).
- 5. *Dormancy:* Canopy is inert, there is no photosynthesis, and autotrophic respiration continues at a low rate. This stage begins when there are no longer leaves in the canopy, and continues until phenological triggers initiate a transition into leaf-out.

The remote sensing community supports the use of five stages for natural vegetation, as time series of satellite data are used to identify key seasonal transitions in vegetation activity between four stages: greenup, maturity, senescence, and dormancy (Hanes et al., 2013; Zhang et al., 2003). After performing sensitivity studies, we separated greenup into two stages (leaf-out and growth) because we found that five stages optimally captures the leaf-out and senescence, while providing sufficient options to simulate carbon flux and pool day-to-day, seasonal, and interannual variability using a mechanistic approach.

2.2. Phenology Stage Selection

The timing and length of each phenology stage is not prescribed, but rather is diagnosed from assimilation rate, climate, day length, leaf pool size, and plant stress from the associated weather and environmental conditions. In order to quantify the shifting between phenology stages, this scheme uses a phenology index (*PI*) such that

$$PI = PS_{GI} \cdot PS_{Wx} \cdot PS_{DayL} \tag{1}$$

where PS_{GI} is a growth index, PS_{Wx} is a weather potential, and PS_{DayL} is a day length potential. PI will be high when it is beneficial to grow new leaves and when the environmental conditions support photosynthesis. In contrast, PI will be 0 either when there is no benefit in growing new leaves or when conditions are not suitable for photosynthesis. To distinguish between the phenology stages, SiB4 applies prescribed threshold values (PI_{Thresh}) to PI.



2.2.1. Growth Index (PS_{GI})

The growth index determines the benefit of growing new leaves as opposed to allocating carbon belowground. We developed this index using the light-use efficiency curve, radiation saturation effect, and climatic sustainability. When the vegetation coverage is low, the addition of each leaf has a significant impact on the photosynthesis rate; however, as the canopy closes and approaches the maximum aboveground biomass that can be supported by the climate, adding more leaves is no longer beneficial. To mimic this behavior, the growth index changes as necessary, starting at its maximum value when the vegetation is minimal (highest potential for growth) and decreasing to its minimal value with a full canopy.

The growth index $({\it PS}_{GI})$ linearly decreases with increasing LAI such that

$$PS_{GI} = \begin{cases} PSG_{Max} & LAI \leq ClimLAI_{Min} \\ \frac{PSG_{Max} - PSG_{Min}}{ClimLAI_{Min} - ClimLAI_{Max}} & ClimLAI_{Min} \leq LAI \leq ClimLAI_{Max} \\ PSG_{Min} & LAI \geq ClimLAI_{Max} \end{cases}$$
(2)

where $ClimLAI_{Min}$ and $ClimLAI_{Max}$ are the diagnosed minimum and maximum LAI for a given climate and PSG_{Max} and PSG_{Min} are parameters.

In order to determine the canopy fullness supported by climatic conditions, this method uses climatological potentials that are 10-year running means initialized during spin-up. $ClimLAI_{Min}$ and $ClimLAI_{Max}$ are calculated such that

$$ClimLAI_{Min} = ClimP^{CL_{C}} + CL_{L} \text{ and}$$
(3)

$$ClimLAI_{Max} = ClimP^{CL_C} + CL_G, \tag{4}$$

where *ClimP* is a climatological suitability index and CL_C , CL_L , and CL_G are parameters.

ClimP can exponentially or linearly scale climatic water availability to cover environmental conditions specific to each vegetation type and is given by

$$ClimP = CP_A (CP_B)^{ClimW} + CP_C (ClimW - CP_D)$$
(5)

$$CP_{Min} \le ClimP \le CP_{Max}$$
 (6)

where *ClimW* is a climatological water availability index and CP_A , CP_B , CP_C , CP_D , CP_{Min} , and CP_{Max} are parameters. *ClimP* is minimal in climates that cannot support vegetation and increases with increasing water availability.

ClimW provides a standardized metric to determine the climatological amount of soil water available to plants, since water availability is a key determinant of plant productivity. For grasslands, *ClimW* is the 10-year running mean of the root-weighted fractional plant available water (*PAW*). *PAW* is defined as

$$PAW = \frac{\sum_{i=1}^{nsoil} RootF_i \left(VL_i - WP \right)}{FC - WP},$$
(7)

where $RootF_i$ is the root fraction in soil layer *i*, VL_i is the volume of liquid water in soil layer *i*, FC is the field capacity, and WP is the wilting point. Since PAW only includes liquid water, this metric is lowered by times and/or layers of soil that are frozen. By averaging PAW over 10 years, ClimW provides a fractional metric for the climatological liquid water obtainable to plants through their roots, with unity representing unrestricted access to water and 0 indicating no water is available.

Relationships between ClimP and ClimW are shown in Figure S1. C4 grasslands use the exponential functionality (using CP_A and CP_B) to rapidly increase ClimP for water availability values in the midrange of ClimW. For these grasslands, the scale of ClimP stretches to values between 0.1 to 2, since small differences in water availability can produce large differences in vegetation greenness and productivity. C3 grasslands use the linear relationship between ClimP and ClimW. Since temperate C3 grasslands cover broader climate conditions, they span the entire range of ClimW and expand the range from one to two. In contrast, arctic grasslands are honed in on the lower ClimW values, since much of the year the ground is frozen.



Using PFT-specific relationships, this method uses site-specific long-term water availability to create climatological minimum and maximum bounds of LAI that can then be compared to current LAI values, determining PS_{GI} . Example PS_{GI} values are shown in Figure S2, where each colored line represents different site-specific *ClimP* values determining the LAI bounds. At the beginning of the growing season when the LAI is minimal, PS_{GI} is at its maximum potential ($PS_{GI,Max}$). As the LAI increases, the benefit of adding more leaves decreases due to increasing respiratory costs or decreasing resources, such as limited light or water. PS_{GI} linearly decreases with increasing LAI until it reaches its minimum value ($PS_{GI,Min}$) when LAI surpasses *ClimLAI_{Max}*. As PS_{GI} decreases, *PI* decreases, advancing the phenology stage.

2.2.2. Weather Potential (PS_{Wx})

The phenology weather potential, PS_{Wx} , utilizes real-time weather conditions and soil moisture to represent the potential photosynthesis. PS_{Wx} is unity at the start of the growing season when environmental conditions are suitable for photosynthesis, ensuring rapid growth immediately at the onset of growth. As the growing season progresses and environmental conditions required for photosynthesis deteriorate, PS_{Wx} decreases, until it reaches zero under highly stressed conditions when photosynthesis is inhibited.

The equation used to calculate PS_{Wx} is

$$PS_{Wx} = \left(\frac{F_{Wx,RM}}{F_{Wx,SM}}\right),\tag{8}$$

where $F_{Wx,RM}$ is combined potential that is sensitive to soil moisture, humidity, and temperature and $F_{Wx,SM}$ is the seasonal maximum of $F_{Wx,RM}$. $F_{Wx,RM}$ is given by

$$F_{Wx,RM} = \frac{1}{2} \left(F_{E,RM} + F_{WA,RM} \right), \tag{9}$$

where $F_{E,RM}$ and $F_{WA,RM}$ are 10-day running mean potentials representing environmental conditions and water availability. Since the overarching purpose of the phenology scheme is to follow the natural plant progression through the growing season, 10-day running means are used to avoid spontaneous reactions in plant growth due to short-lived weather systems while still responding to synoptic and seasonal-scale anomalies. The running mean length is an input parameter and can be changed as the user sees appropriate; however, SiB4 uses a 10-day running mean for all grasslands, and sensitivity studies justifying this choice are provided in section S1.

 F_E is a combined potential from three environmental conditions, such that

$$F_E = F_{LH} F_{RZ} F_T.$$
⁽¹⁰⁾

All three conditions are potentials that equal unity under optimal conditions and 0 in adverse environmental conditions (Sellers, 1987). The F_{LH} potential is a function of water vapor pressure and represents leaf humidity stress (Sellers et al., 1992). The F_{RZ} potential represents root-zone water stress and can be summarized as follows: if the total column soil moisture is above the field capacity, then there is maximum potential and no stress; in contrast, if the soil moisture is below the wilting point, then the plant is totally stressed and thus has zero potential (Baker et al., 2008). The F_T factor is a function of temperature and incorporates high- and low-temperature stress (Baker et al., 2008; Jarvis, 1976).

To intensify the effects of water availability, $F_{WA,RM}$ is the 10-day running mean of the root-weighted fraction of *PAW* in the top three soil layers (*PAW*_{Top}), which is given by

$$PAW_{Top} = \frac{\sum_{i=1}^{3} RootF_i \left(VL_i - WP \right)}{RootF_3 (FC - WP)} \quad \text{and} \tag{11}$$

$$RootF_3 = \sum_{i=1}^{3} RootF_i,$$
(12)

where $RootF_3$ is the total rooting fraction in the top three soil layers.



2.2.3. Day Length Potential (*PS*_{DavL})

Photoperiod is a dominant cue to initiate senescence processes in the arctic (Ernakovich et al., 2014; Estiarte & Peñuelas, 2015; Pau et al., 2011). To progress the phenology as days get shorter, as well as to prohibit leaf-out during mild weather events during the winter, this strategy uses a photoperiod potential (PS_{DayL}). As the days shorten, PS_{DayL} decreases, lowering *PI* and advancing the phenology stage. The equations for PS_{DayL} are

$$PS_{DayL} = \begin{cases} 1 - PSD_{Mul} \left(DayL - PSD_{Ref} \right) & DayL_{dt} \ge 0\\ 1 - DayL_0 - PSD_{Mul} \left(1 - DayL \right) & DayL_{dt} \le 0 \end{cases}$$
(13)

$$PS_{DayL} \ge PSD_{Min} \tag{14}$$

$$DayL_{O} = PSD_{Mul} \left(DayL_{Max} - PSD_{Ref} \right)$$
⁽¹⁵⁾

where PSD_{Min} , PSD_{Mul} , and PSD_{Ref} are parameters specifying the minimum, daily coefficient, and reference value for PS_{DayL} , respectively. DayL is the length of day, $DayL_{dt}$ is the change in the length of day, $DayL_{Max}$ is the length of the longest day of the year, and $DayL_O$ is an offset value.

This set of equations allows the strength of the photoperiod phenological response to vary with vegetation type and with latitude. For PFTs that exist across all latitudes, the length of day can have no impact on the phenology at low latitudes (e.g., tropics) while the length of day can have significant phenological impacts at high latitudes (e.g., arctic). Examples of PS_{DavL} at 45°N are shown in Figure S4.

2.3. Growing Season Start

Unlike crops that have a definitive growing season that begins with a planting date, the beginning of the growing season for grasslands is not as straightforward and depends on climate: temperate grasslands may have strong seasonality with a dormancy period during the winter, while tropical grasslands may continue growing all year. Adding even more complexity, temperate grasslands may have two growing periods within a single year, with a brief period of browning in-between when the environmental conditions are too hot and dry to support growth. To mimic natural behavior, the phenology scheme needs to be dynamic, with the capacity to return to the leaf-out or growing stages when necessary, while not triggering leaf-out midseason for vegetation with a clearly defined single seasonal cycle upon experiencing midseason stress.

To accomplish this, one of the phenological potentials, PS_{Wx} can be reset in order to help obtain the leaf-out stage when warranted. With PS_{Wx} being a ratio, it is reset to unity by setting the seasonal maximum value $F_{E,SM}$ equal to the current value $F_{E,RM}$. For seasonal vegetation, this ensures rapid growth from a minimal leaf pool when day lengths are not substantially decreasing. However, for nonseasonal vegetation that may not ever obtain the conditions required for resetting this potential, leaf-out or growth can still be obtained if the conditions support rapid growth.

To determine when $F_{E,SM}$ is reset, SiB4 uses two flags. The first is the growing season suitability flag (*FlagG*), which is true when conditions are suitable for photosynthesis. Following the growing season start drivers used by Jolly et al. (2005), SiB4 uses temperature, soil moisture, and day length requirements to begin a growing season. For temperature, the daily mean temperature (TM_D) must be greater than a specified temperature (parameter FGT_{Min}) for a specific number of days (parameter FGT_{Len}). For soil moisture, SiB4 uses the root-weighted total available fraction of water in the top three soil layers (TAW_{Top}), which is given by

$$TAW_{Top} = \frac{\sum_{i=1}^{3} RootF_i \left(\left(VL_i + VI_i \right) - WP \right)}{RootF_3 (FC - WP)} \quad \text{and} \tag{16}$$

where VI_i is the volume of ice in soil layer *i*. Whereas *PAW* is penalized during time periods of frozen soil, *TAW* includes both liquid and ice water contributions and thus is a metric for total water availability. The top three soil layers (26 mm depth) contain ~75% of simulated grassland roots; thus, these layers can be used to indicate water availability while responding quickly to precipitation events given their relatively shallow depth. To start the growing season, TAW_{Top} must be above a specified value (parameter FGW_{Min}) for a specific number of days (parameter FGW_{Len}). For light, the day length must be longer than a specified minimum length. When the days are getting longer, the required day length is given by the parameter FGD_{MinI} .

When the days are getting shorter, the required day length is given by the parameter FGD_{MinD} . If all three of these conditions are met, then FlagG is true, otherwise FlagG is false.

The second flag used to determine the start of the growing season is an assimilation flag (*FlagA*), which is set to true when the assimilation rate becomes minimal at the end of the growing season. The assimilation flag makes use of an assimilation potential (F_A) and is true when

$$F_A < Flag A_{Thresh}$$
 and (17)

$$F_A = \frac{AssimD_{RM}}{AssimD_{SM}},\tag{18}$$

where $FlagA_{Thresh}$ is a parameter, $AssimD_{RM}$ is a 10-day running mean of the daily mean assimilation rate AssimD, and $AssimD_{SM}$ is the seasonal maximum of $AssimD_{RM}$. $AssimD_{SM}$ is reset to $AssimD_{RM}$ at the beginning of the growing season, when $F_{E,SM}$ is also reset. The occurrence of both FlagA and FlagG indicates that the vegetation can commence leaf-out to start a new growing season.

ŀ

2.4. Phenology and Physiology Interactions

We hypothesize that two physiological processes are directly dependent on phenophase: carbon allocation and leaf photosynthetic capacity. Carbon allocation is a critical physiological process that determines the fate of assimilated carbon and affects plant growth (Xia et al., 2017). A compilation of research suggests that allocation to the leaf pool decreases with advancing phenology stage, with a shift in allocation from aboveground to belowground through the growing season (Aguado et al., 2016; De Kauwe et al., 2014; Guillemot et al., 2017; Ise et al., 2010; Jones et al., 2013; Klein and Hoch, 2015; Negrón-Juárez et al., 2015). This progression makes sense intuitively as well: at the beginning of the growing season when the leaf pool is small, plants are limited by the light they collect and thus allocate resources to growing leaves. At the end of the season plants allocate more resources belowground to the roots for two reasons: (1) in many ecosystems water becomes a limiting resource (e.g., Blair et al., 2014; Rishmawi et al., 2016; Sala et al., 1988) and (2) roots store carbon required for growth at the beginning of the next growing season.

Using the philosophy that plants adjust their allocation to improve their most limiting resource, SiB4 includes seasonally varying allocation by including stage-specific allocation fractions (parameter $AllocP_{nlpool, nstage}$) that specify the fraction of photosynthesized carbon allocated to each live pool for each phenology stage. Allocation to the leaf pool in the case of light limitation is captured in the phenology stage selection: if the environmental conditions support growth and if adding leaves would provide an improvement to photosynthesis at a reasonable cost, then the phenology stage shifts to the leaf-out or growth stages with high leaf allocation fractions. Moisture sensitivity is also included in the phenology stage determination: if the vegetation is moisture stressed, then the phenophase will shift toward later stages that have more allocation to the roots. While the phenophase captures allocation changes throughout the season, complimentary weather-based adjustments provide a way to capture day-to-day variability, with adjustments being made based on water and temperature conditions (Friedlingstein et al., 1999; Schaefer et al., 2008).

Second, plants can change their leaf photosynthetic capacity (maximum carboxylation rate of rubisco, V_{Max}) throughout the season. V_{Max} is an important control on the photosynthetic rate, changing the amount of carbon available for growth and maintenance and having significant impacts on the carbon cycle, energy exchanges, and hydrology (Baldocchi et al., 2001; Bauerle et al., 2012; Groenendijk et al., 2011; Wilson & Baldocchi, 2000). Increasing evidence indicates that large seasonal variations in V_{Max} occur in terrestrial ecosystems (Bauerle et al., 2012; Groenendijk et al., 2011; Houborg et al., 2013; Zhou et al., 2014); and specifying V_{Max} by phenophase ties it to seasonal development and provides a mechanistic way to change it throughout the growing season. V_{Max} increases through the growth stage, then decreases as the leaves age (Restrepo-Coupe et al., 2013; Saleska et al., 2014; Suzuki et al., 2010; Wu et al., 2016; Xu et al., 2017).

Implementing prognostic phenology using dynamic stages provides a powerful mechanism to integrate ecosystem processes, linking carbon pools (storage), land-atmosphere (energy) fluxes, and biosphere-atmosphere (carbon) exchanges. Phenophases can be determined from environmental conditions and used to connect physiology and biogeochemistry, and Figure 1 provides an illustration of this integration. Using phenophase to change plant physiology shifts the focus from prescribing physical properties, which are highly variable in space and time, to predicting them. Rather than using empirical allometric relationships, ecosystem characteristics emerge from processes that vary with weather and climate. For example,





Figure 1. Phenology-flux-pool integration.

rather than using a predefined root-to-shoot ratio, this is an emergent property in this model, which is bounded by climate and weather interactions rather than by specified parameters.

3. Senescence

3.1. Turnover

Similar to SiB-CASA (Schaefer et al., 2008), pool turnover is tied to respiration; however, SiB4 introduces new scaling coefficients that are described in this section. To capture the loss of carbon from any live pool (*lp*) that is transferred to a dead carbon pool, SiB4 uses the equation

$$LossT_{lp,t} = \frac{(1 - E_{lp})M_{A,t}M_{F,t}M_{H,t} C_{lp}}{\tau_{lp}}dt,$$
(19)

where $Loss T_{lp,t}$ is the loss due to pool turnover at time step t, E_{lp} is the pool respiration efficiency, C_{lp} is the pool carbon, τ_{lp} is the pool turnover time, and dt is the time step length. $M_{A,t}$, $M_{F,t}$, and $M_{H,t}$ are the assimilation rate, freeze inhibition, and high temperature scaling coefficients at time step t, respectively. The base turnover time is the longest for the product pool based on the premise that seeds and flowers are carbon intensive to

produce yet require less energy to maintain, and it is the shortest for leaves due to the high maintenance cost of rubisco (González-Zurdo et al., 2016; Suzuki et al., 2010).

For grasslands, we hypothesize that aboveground biomass mortality occurs primarily at the end of the growing season as litterfall (see next section); thus, for the canopy pools $E_{cp} = 0.95$. In contrast, root mortality depends solely on turnover given that root turnover is still poorly constrained due to the complexity of factors that influence root mortality, the heterogeneity in root life spans, and difficulties in measurement techniques (Urrutia-Jalabert et al., 2017; Riley et al., 2009; Smithwick et al., 2014). Since root production, respiration, mortality, and turnover all interact, SiB4 uses root respiration efficiencies (E_{rp}) ranging from 0.45 to 0.6.

Studies have suggested that respiration scales predominantly with the assimilation rate (Flexas et al., 2006; Meir et al., 2008; Molchanov, 2009). SiB4 introduces an assimilation rate scaling coefficient (M_A) that is calculated separately for the canopy and root pools. For any canopy pool (cp), M_A is calculated as

$$M_{A,cp} = \begin{cases} CA_{Min} & AssimD < CRT_{AL} \cdot ClimA \\ \frac{CA_{Max} - CA_{Min}}{CA_H - CA_L} & CA_L \cdot ClimA \le AssimD \le CA_H \cdot ClimA \\ CA_{Max} & AssimD > CA_H \cdot ClimA \end{cases}$$
(20)

where *ClimA* is the 10-year running mean of *AssimD*. CA_{Min} , CA_{Max} , CA_L , and CA_H are parameters. The assimilation rate scaling coefficient for the roots $(M_{A,rp})$ uses the same formulation, but with parameters specific to the root pools $(RA_{Min}, RA_{Max}, RA_L, \text{ and } RA_H)$.

SiB4 uses two temperature processes to adjust the carbon pool turnover rates. First, freezing temperatures inhibit microbial activity slowing root respiration (Hobbie et al., 2000; Mikan et al., 2002; Schaefer et al., 2008). SiB4 exponentially decreases the freeze inhibition scaling coefficient (M_F) for any canopy pool *cp* using the equations

$$M_{F,cp} = e^{CF_{Mul}(TC - CF_{Ref})} \quad \text{and} \tag{21}$$

$$CF_{Min} \le M_{F,cp} \le 1,\tag{22}$$

where *TC* is the canopy temperature and CF_{Mul} , CF_{Ref} , and CF_{Min} are parameters. This same formulation holds for the roots ($M_{F,rp}$) using root-specific parameters (RF_{Mul} , RF_{Ref} , and RF_{Min}), with $M_{F,rp}$ being calculated at every soil layer containing roots using the corresponding soil temperature (*TD*).

Second, studies have shown autotrophic respiration and root mortality increases with temperature (Bai et al., 2010; Gill and Jackson, 2000; McCormack et al., 2014). As in previous versions of SiB (Baker et al., 2008; Denning et al., 1996; Schaefer et al., 2008), the high temperature scaling coefficient (M_H) use a Q10 relationship; however, in SiB4 M_H is calculated separately for the canopy and root pools using the separate reference temperatures (CT_{Ref} , RT_{Ref}), Q10 bases (CT_{Q10} , RT_{Q10}), and maximum contributions (CT_{Max} , RT_{Max}).



3.2. Litterfall

Using phenology stages updated daily provides a mechanistic approach for modeling the start of season and the progression throughout the growing season; however, changing allocation and photosynthetic rate alone combined with continuous turnover is not enough to capture the rapid grassland browning seen at the end of the growing season. To predict senescence, this approach adds three explicit processes for litterfall.

Day length, temperature, and soil moisture are commonly used environmental cues for senescence (Borchert et al., 2005; Dahlin et al., 2015; Jolly et al., 2005; Kim et al., 2015; White et al., 1997). The transfer of biomass in canopy carbon pools (*cp*) to dead carbon pools is calculated as

$$LossL_{cp,t} = (T_{DL} + T_F + T_W)C_{cp}dt,$$
(23)

where *LossL* is the canopy pool loss of carbon at time step t, T_{DL} is the fraction of the canopy pool that is transferred to the dead carbon pools due to decreasing day length, T_F is the fraction of the canopy pool that is transferred due to freezing temperatures, and T_W is the fraction of the canopy pool that is transferred due to water deprivation.

For day length, T_{DL} increases as the days get shorter following

$$T_{DL} = LTD_C (DayL_{Max} - DayL)(DayL_{Max} - LTD_R) \quad \text{and}$$
(24)

$$0 \le T_{DL} \le LTD_{Max} \tag{25}$$

where LTD_C , LTD_R , and LTD_{Max} are parameters. An example of daylength transfer fractions calculated at 45°N is shown in Figure S5.

For freezing temperatures, T_F increases with decreasing temperatures using a Q10 relationship following

$$T_F = LTF_{Q10}^{0.01(LTF_{Ref} - TC)} - 1 \quad \text{and}$$
(26)

$$0 \le T_F \le LTF_{Max} \tag{27}$$

where LT_{Fa10} , LT_{Fmax} , and LT_{Fref} are parameters. An example of T_F is shown in Figure S6.

For moisture deprivation, T_W increases with lowering *PAW* such that

$$T_W = LTW_C(LTW_B^{10(PAW-LTW_R)} - 1) \quad \text{and} \tag{28}$$

$$0 \le T_W \le LTW_{Max} \tag{29}$$

where LTW_B , LTW_C , LTW_R , and LTW_{Max} are parameters. An example of T_W is shown in Figure S7.

4. Grazing

Grasslands have a long history of grazing, and grazing is an important disturbance that has direct impacts on the carbon cycle from local-scale vegetation dynamics, to regional productivity, to global sources and sinks (Kang et al., 2007; Knapp et al., 2012; Koerner et al., 2014; Yan et al., 2013). Grazing removes aboveground plant material, which has considerable effects on the land-atmosphere interactions by changing the aboveground vegetation cover, and hence altering the LAI, litter, albedo, roughness length and resulting energy and momentum exchanges. Changing the aboveground carbon pools also alters the carbon fluxes and plant productivity, which feeds back to impact not only the root biomass, but also the resulting carbon stored in the soil. Extensive grazing occurs globally, and these local impacts scale up to have global implications.

Since grazing has widespread impacts, SiB4 includes a rudimentary grazing scheme. Following grassland management practices for sustainable grazing, the amount of removed carbon is a constant fraction of the long-term aboveground biomass (which for grasslands corresponds to the net primary production, NPP). While grazing estimates are improving, grazing intensity remains highly variable and uncertain, ranging from none to 100% in overgrazed pastures (Fetzel et al., 2017). SiB4 uses two different grazing intensities depending on climatological (10-year running mean) LAI, both of which are adjustable parameters. Currently, for grasslands with climatological maximum LAI \geq 1, 40% of annual NPP is grazed, corresponding to moderate grazing practices. Otherwise, 10% of annual NPP is grazed. Grazing occurs daily once grasslands have a non-minimal LAI, and grazed carbon is primarily transferred to the dead carbon pools, with a small fraction immediately respired. Details for the grazing scheme are in the supplemental material (section S2).





Figure 2. SiB4 structural diagram. Green boxes show the live carbon pools, and tan boxes show the dead carbon pools. The flow of carbon is shown in the black lines. The release of carbon into the atmosphere is indicated by the green lines for autotrophic respiration and by the burgundy lines for heterotrophic respiration. The yellow boxes show the predominant influences on the carbon pools, and the white boxes show the land-atmosphere exchanges of carbon.

5. Terrestrial Carbon Cycle Prediction

Prognostic phenology provides a powerful mechanism to integrate ecosystem processes, linking carbon pools (storage), land-atmosphere (energy) fluxes, and biosphere-atmosphere (carbon) exchanges and leading to a fully predictive terrestrial carbon cycle. Rather than relying on satellite data for the vegetation state, SiB4 determines this from fully prognostic carbon pools. Figure 2 shows an overview of the carbon pools in SiB4, along with the processes controlling their fluxes and transfers. Above and belowground live biomass and dead carbon pools can be predicted from a combination of phenophase-dependent assimilation rates and carbon allocation with environmentally dependent turnover rates, and these pools and resulting vegetation states feed back to impact land surface properties and radiative transfer, carbon and energy fluxes, and hydrology.

To combine these processes, every time step SiB4 computes the albedo, radiation, temperature, and soil moisture, along with the resulting energy exchanges, moisture fluxes, and carbon fluxes. Every day SiB4 sums the carbon fluxes and calculates carbon transfers required from disturbance (grazing). Once daily, the gains and losses are combined to update the carbon pools and phenology stage. Using the updated pools, all related land surface properties are diagnosed and used for subhourly photosynthetic assimilation as well as subhourly autotrophic and heterotrophic respiration and pool transfer rates. This sequence completes the carbon cycle, providing self-consistent predicted vegetation state, carbon pools, and land-atmosphere exchanges. An illustration of the timing sequence for the SiB4 carbon cycle along with a step-by-step outline of the calculation sequence is in section S3.

6. Site Demonstration

For an in-depth analysis of grassland carbon cycling using SiB4, eddy-covariance carbon fluxes, remotely-sensed vegetation states, and *in situ* net primary productivity data, please see Part 2 (Haynes et al., 2019). Here, we walk through one year at a single site to demonstrate the new methodology.







Figure 3. Temperate grassland seasonal cycle (US-Kon, 39.1° N, -96.56° E, 2002). (Top left) GPP (green), NPP (blue), R_A (orange), and R_H (red). (Top right) Total phenology index (black), growth potential (green), weather potential (red), and day length potential (blue). The threshold values that determine the phenology stages are shown in the dashed black lines, and the phenology stages are indicated in brown. (Bottom right) Allocation fractions for the leaf pool (green), stem pool (purple), product pool (blue), fine roots (red), and coarse roots (orange). (Bottom left) Live carbon pools (colors, left axis) and LAI (black, right axis). The MODIS LAI is shown in the dashed black line.

The predicted carbon cycle at a Northern Hemisphere temperate grassland site (US-Kon, $39.08^{\circ}N$, $-96.56^{\circ}E$, 2002, Knapp et al., 1998) is shown in Figure 3. During 2002, this site has two peaks in carbon fluxes and aboveground biomass to show how the phenology scheme is dynamic and capable of reacting to ecological and environmental conditions. The panels are organized in a manner to illustrate the daily calculation sequence.

Daily mean carbon fluxes (GPP, NPP, R_A , and R_H) are shown in the top left panel of Figure 3. This panel shows the seasonal uptake and respiration of carbon, as well as the day-to-day variability in the fluxes. During the growth and mature leaf stages, R_A is approximately half of GPP, and during the dormancy stage the maintenance costs exceed the carbon being taken up from photosynthesis due to the difficult environmental conditions.

The phenology index and the three contributing potentials are shown in the upper right panel of Figure 3. At the beginning of the year during winter, the total phenology index is near 0, since cold temperatures force the weather potential to low values. Since the LAI is minimal, the growth potential is at its maximum value; and since the days are getting longer, the day length potential is also at its maximum. In spring when the conditions become suitable for photosynthesis, the weather potential jumps up to unity, causing a maximum value in the phenology index, resulting in leaf-out. With the growth and day length indices still at their maximum values, the phenology index follows the weather potential, and the phenophase alters between leaf-out, growth, and maturity depending on daily conditions. Toward the middle of the year, the LAI becomes large enough that the growth index begins decreasing. At the same time, the site experiences heat and water stress, causing the weather potential to decrease and shifting the phenophase into dormancy. After several weeks, the environmental conditions improve, causing the weather potential to jump up to unity; however, at this



late point in the season the day length index is decreasing, which limits the phenophase to the growth stage. As the year progresses and conditions deteriorate, both the weather potential and day length index decrease, causing the phenology to advance to the stress and dormancy stages. The last few days of the year after the solstice when the days begin getting longer, the day length index jumps back to unity to prepare for another growing season, while the phenology index remains low due to the unsuitable environmental conditions and low values of the weather potential.

The allocation fractions calculated from combining the phenology fractions with environmental adjustments are shown in the bottom right panel of Figure 3. The figure shows the pattern of leaf allocation decreasing throughout the season as more carbon is allocated to the roots. This panel illustrates the variability in carbon allocation and shows that although the main seasonal phenological progression dominates the allocation, there are contributions from the environmental adjustments, especially at the end of the growing season.

Finally, from all of the daily gains and losses, the carbon pools are updated. Sample seasonal cycles of the grassland live carbon pools are shown in the lower left panel of Figure 3. As the growing season progresses, the live carbon pools change daily. During the first half of the growing season, the above ground carbon pools grow, until the vegetation becomes mature and summer stresses begin. When this occurs, the root pool grows while the aboveground pools decrease. Once the site becomes stressed, the aboveground pools decrease both from increasing maintenance costs and from increasing carbon transfer to the dead carbon pools. In the second half of the year, the return of the growth stage from more optimal growing conditions causes a secondary peak in the aboveground carbon pools. At the end of the growing season, as the GPP continues to decrease, the costs of the aboveground live pool begin to outweigh the carbon being provided, and this combined with the high transfer rate due to stresses cause a rapid decrease in the aboveground pools.

The lower left panel of Figure 3 also shows the SiB4 LAI (black line) compared to MODIS LAI (dashed black line) from the MODIS Land Product Subsets Project (http://daac.ornl.gov/MODIS/). The MODIS values shown are the mean between two products: MODIS Terra Leaf Area Index 8-Day Composite (MOD15A2) (ORNL DAAC, 2008a) and the equivalent MODIS Aqua product (MYD15A2) (ORNL DAAC, 2008b). The observed and predicted phenology show similar patterns: the grassland remains dormant until mid-April, rapidly greens-up to reach a maximum in June, browns-down as the grassland experiences water and temperature stress in July, greens-up again in September due to late-season warm temperatures and rainfall in September, then finally senesces and enters dormancy in late October. However, SiB4 begins the growing season too early at this site, has too broad of a primary growing season, and senesces too rapidly at the end of the season compared to MODIS. Using the dynamic phenostage approach, SiB4 is able to simulate both the maximum LAI as well as the secondary late-season peak, in addition to the rapid transitions between these phenological events.

7. Conclusions

Since phenology is an important yet difficult component of models, we developed an innovative predictive phenology method that continuously predicts vegetation phenophase to link biophysical, biogeochemical, and phenological processes. The phenophase determines carbon allocation, maximum photosynthesis rate, and biomass transfer, providing a method to seasonally vary these important quantities. This dynamic approach closes the terrestrial carbon cycle using environmental mechanisms. Integrating phenology, carbon fluxes, and biomass provides a powerful constraint on the terrestrial carbon cycle, leading to an emergence of ecosystem characteristics, behaviors, and properties that can be used for evaluation, rather than relying on empirical or tuned parameters. For example, the root-shoot ratio in SiB4 is not prescribed, but instead emerges as a long-term property from daily changing carbon allocation. By viewing root-shoot ratios as emergent properties from different patterns in carbon allocation, evaluating this metric yields insight into allocation not only for different vegetation types and for different regions, but also for ecosystem responses to climate and perturbations in either time or space.

A full understanding of the linkage between phenology and terrestrial biophysics is unknown, particularly on community and ecosystem scales, thus using phenophases to link phenology with terrestrial biology provides an approach that can be used to learn about these relationships. Combining prognostic phenology with established sub-hourly photosynthesis, radiation, and hydrology processes, SiB4 is a tool that can be used to



yield predictions across times scales ranging from minutes to seasons to centuries. A wide branch of research is using phenology to learn about biological mechanisms, ecological processes, and climate change, and SiB4 can be included in these studies to investigate the impact of changing environments on ecosystems, including impacts on land-atmosphere fluxes, live biomass, and carbon pools. As with any model, future work is required to improve SiB4; however, it is our hope that model improvement will coincide with learning about the processes involved in land-atmosphere interactions, the terrestrial carbon cycle, and climate change.

Evaluation at a grassland site illustrated the feasibility of this method, and a complimentary paper further evaluates this approach across grassland sites worldwide (Haynes et al., 2019). SiB4 has also been used to investigate the impact of drought in temperate grasslands across the Great Plains of North America (Curry, 2016), to quantify land surface behavior across the Sahel (Smith et al., 2018), and to investigate phenology using solar induced fluorescence (Cheeseman, 2018).

References

- Aguado, P., Curt, M., Pereira, H., & Fernández, J. (2016). The influence of season on carbon allocation to suberin and other stem components of cork oak saplings. *Tree Physiology*, 37, 165–172. https://doi.org/10.1093/treephys/tpw116
- Bai, W., Wan, S., Niu, S., Liu, W., Chen, Q., Wang, Q., et al. (2010). Increased temperature and precipitation interact to affect root production, mortality, and turnover in a temperate steppe: Implications for ecosystem C cycling. *Global Change Biology*, 16, 1306–1316. https://doi. org/10.1111/j.1365-2486.2009.02019.x
- Baker, I., Prihodko, L., Denning, A., Goulden, M., Miller, S., & da Rocha, H. (2008). Seasonal drought stress in the Amazon: Reconciling models and observations. *Journal of Geophysical Research*, 113, G00B01. https://doi.org/10.1029/2007JG000644
- Baldocchi, D., Falge, E., Gu, L., Olson, R., Hollinger, D., Running, S., et al. (2001). FLUXNET: A new tool to study the temporal and spatial variability of ecosystem-scale carbon dioxide, water vapor, and energy flux densities. *Bulletin of the American Meteorological Society*, *82*(11), 2415–2435.
- Bauerle, W. L., Oren, R., Way, D. A., Qian, S. S., Stoy, P. C., Thornton, P. E., et al. (2012). Photoperiodic regulation of the seasonal pattern of photosynthetic capacity and the implications for carbon cycling. *Proceedings of the National Academy of Sciences*, *109*(22), 8612–8617. https://doi.org/10.1073/pnas.1119131109
- Blair, J., Nippert, J., & Briggs, J. (2014). Grassland ecology. In R. Monson (Ed.), *Ecology and the Environment, the Plant Sciences* (Vol. 8, pp. 389–423). New York, NY: Springer. https://doi.org/10.1007/978-1-4614-7501-9_14
- Borchert, R., Robertson, K., Schwartz, M., & Williams-Linera, G. (2005). Phenology of temperate trees in tropical climates. International Journal of Biometeorology, 50, 57–65. https://doi.org/10.1007/s00484-005-0261-7
- Caldararu, S., Purves, D. W., & Palmer, P. I. (2014). Phenology as a strategy for carbon optimality: A global model. *Biogeosciences*, 11(3), 763–778. https://doi.org/10.5194/bg-11-763-2014
- Cheeseman, M. (2018), Productivity and phenology in a process-driven carbon cycle model, Master's thesis, Colorado State University. Chen, X. (2003). East Asia. In M. Schwartz (Ed.), *Phenology: An integrative environmental science* (pp. 11–26). Dordrecht, The Netherlands:
- Kluwer Academic Publishers. Chuine, I., Garcia de Cortazar-Atauri, I., Dramer, K., & Hanninen, H. (2013). Plant development models. In M. Schwartz (Ed.), *Phenology: An integrative environmental science* (2nd ed., pp. 275–294). The Netherlands: Dordrecht.
- Curry, R. (2016), Impacts of drought on grassland productivity across the wet-dry gradient in the U.S. Great Plains in 2010-2012. Master's thesis, Colorado State University, Fort Collins, Colorado, USA.
- Dahlin, K. M., Fisher, R. A., & Lawrence, P. J. (2015). Environmental drivers of drought deciduous phenology in the Community Land Model. *Biogeosciences*, 12(16), 5061–5074. https://doi.org/10.5194/bg-12-5061-2015
- Dahlin, K. M., Ponte, D. D., Setlock, E., & Nagelkirk, R. (2017). Global patterns of drought deciduous phenology in semi-arid and savanna-type ecosystems. *Ecography*, 40(2), 314–323. https://doi.org/10.1111/ecog.02443
- De Kauwe, M. G., Medlyn, B. E., Zaehle, S., Walker, A. P., Dietze, M. C., Wang, Y.-P., et al. (2014). Where does the carbon go? A model-data intercomparison of vegetation carbon allocation and turnover processes at two temperate forest free-air CO₂ enrichment sites. *New Phytologist*, 203(3), 883–899. https://doi.org/10.1111/nph.12847
- Denning, A., Collatz, G., Zhang, C., Randall, D., Berry, J., Sellers, P., et al. (1996). Simulations of terrestrial carbon metabolism and atmospheric CO₂ in a general circulation model. Part 1: Surface carbon fluxes. *Tellus*, *48*, 521–542.
- Eamus, D., Huerte, A., & Yu, Q. (2016). Vegetation dynamics: A synthesis of plant ecophysiology, remote sensing, and modelling. Cambridge, U.K.: Cambridge University Press. https://doi.org/10.1017/CBO9781107286221
- Ernakovich, J. G., Hopping, K. A., Berdanier, A. B., Simpson, R. T., Kachergis, E. J., Steltzer, H., & Wallenstein, M. D. (2014). Predicted responses of arctic and alpine ecosystems to altered seasonality under climate change. *Global Change Biology*, 20(10), 3256–3269. https://doi.org/10.1111/gcb.12568
- Estiarte, M., & Peñuelas, J. (2015). Alteration of the phenology of leaf senescence and fall in winter deciduous species by climate change: Effects on nutrient proficiency. *Global Change Biology*, *21*(3), 1005–1017. https://doi.org/10.1111/gcb.12804
- Fetzel, T., Havlik, P., Herrero, M., Kaplan, J. O., Kastner, T., Kroisleitner, C., et al. (2017). Quantification of uncertainties in global grazing systems assessment. Global Biogeochemical Cycles, 31, 1089–1102. https://doi.org/10.1002/2016GB005601
- Fisher, J., Richardson, A., & Mustard, J. (2007). Phenology model from surface meteorology does not capture satellite-based greenup estimations. *Global Change Biology*, 13(3), 707–721. https://doi.org/10.1111/j.1365-2486.2006.01311.x
- Flexas, J., Bota, J., Galmés, J., Medrano, H., & Ribas-Carbó, M. (2006). Keeping a positive carbon balance under adverse conditions: Responses of photosynthesis and respiration to water stress. *Physiologia Plantarum*, 127(3), 343–352. https://doi.org/10.1111/j. 1399-3054.2006.00621.x
- Forkel, M., Carvalhais, N., Schaphoff, S., Bloh, W. V., Migliavacca, M., Thurner, M., & Thonicke, K. (2014). Identifying environmental controls on vegetation greenness phenology through model-data integration. *Biogeosciences*, 11(23), 7025–7050. https://doi.org/10.5194/ bg-11-7025-2014

Acknowledgments

SiB4 is available online (https://gitlab. com/kdhaynes/sib4_corral), and permissions will gladly be provided upon contacting the corresponding author. We would like to thank Sarah Gallup for her helpful draft revisions and suggestions. We also would like to thank the anonymous reviewers for their helpful comments. The National Aeronautics and Space Administration (NASA) funded this research with contributions from four contracts: Terrestrial Ecology, NNX11AB87G; Carbon Monitoring System Program, NNX12AP86G; Science Team for the OCO-2 Missions, NNX15AG93G; and Carbon Cycle Science, NNX14AI52G. The Department of Energy ProjectDE-SC00014438 also contributed funding to this project.

Friedl, M., Gray, J., Melaas, E., Richardson, A., Hufkens, K., Keenan, T., et al. (2014). A tale of two springs: Using recent climate anomalies to characterize the sensitivity of temperate forest phenology to climate change. *Environtal Research Letters*, 9, 054006. https://doi.org/ 10.1088/1748-9326/9/5/054006

Friedlingstein, P., Joel, G., Field, C., & Fung, I. (1999). Toward an allocation scheme for global terrestrial carbon models. *Global Change Biology*, 5, 755–770.

- Fu, Y., Zhang, H., Dong, W., & Yuan, W. (2014). Comparison of phenology models for predicting the onset of growing season over the northern hemisphere. PLoS ONE, 9(10). https://doi.org/10.1371/journal.pone.0109544
- Gill, R., & Jackson, R. (2000). Global patterns of root turnover for terrestrial ecosystems. Net Phytol., 147, 13–31.
- González-Zurdo, P., Escudero, A., Babiano, J., García-Ciudad, A., & Mediavilla, S. (2016). Costs of leaf reinforcement in response to winter cold in evergreen species. *Tree Physiology*, 36(3), 273–286. https://doi.org/10.1093/treephys/tpv134
- Groenendijk, M., Dolman, A. J., Ammann, C., Arneth, A., Cescatti, A., Dragoni, D., et al. (2011). Seasonal variation of photosynthetic model parameters and leaf area index from global fluxnet eddy covariance data. *Journal of Geophysical Research*, 116, G04027. https://doi.org/ 10.1029/2011JG001742
- Gu, L., Post, W., Baldocchi, D., Black, T., Verma, S., Vesala, T., & Wofsy, S. (2003). In M. Schwartz (Ed.), Phenology of vegetation photosynthesis, in Phenology: An integrative environmental science (p. 566). The Netherlands: Dordrecht.
- Guillemot, J., Francois, C., Hmimina, G., Dufrêne, E., Martin-StPaul, N., Soudani, K., et al. (2017). Environmental control of carbon allocation matters for modelling forest growth. *New Phytologist*, 214(1), 180–193. https://doi.org/10.1111/nph.14320
- Hanes, J., Richardson, A., & Klosterman, S. (2013). Mesic temperate deciduous forest phenology. In M. Schwartz (Ed.), Phenology: An integrative environmental science, (2nd ed., pp. 211–224). The Netherlands: Dordrecht.
- Haynes, K., I. Baker, A. Denning, S. Wolf, G. Wohlfahrt, G. Kiely, and R. Minaya (2019), Representing grasslands using dynamic prognostic phenology based on biological growth stages: Part 2. Carbon cycling, Journal of Advances in Modeling Earth Systems.
- Hobbie, S., Schimel, J., Trumbore, S., & Randerson, J. (2000). Controls over carbon storage and turnover in high-latitude soils. *Global Change Biology*, *6*, 196–210.
- Houborg, R., Cescatti, A., Migliavacca, M., & Kustas, W. (2013). Satellite retrievals of leaf chlorophyll and photosynthetic capacity for improved modeling of GPP. Agricultural and Forest Meteorology, 177, 10–23. https://doi.org/10.1016/j.agrformet.2013.04.006
- Ise, T., Litton, C. M., Giardina, C. P., & Ito, A. (2010). Comparison of modeling approaches for carbon partitioning: Impact on estimates of global net primary production and equilibrium biomass of woody vegetation from MODIS GPP. Journal of Geophysical Research, 115, G04025. https://doi.org/10.1029/2010JG001326
- Jarvis, P. (1976). The interpretation of the variations in leaf water potential and stomatal conductance found in canopies in the field. *Philosophical Transactions of the Royal Society B*, 273, 593–610.
- Jolly, W. M., Nemani, R., & Running, S. W. (2005). A generalized, bioclimatic index to predict foliar phenology in response to climate. Global Change Biology, 11(4), 619–632. https://doi.org/10.1111/j.1365-2486.2005.00930.x
- Jones, C., Robertson, E., Arora, V., Friedlingstein, P., Shevliakova, E., Bopp, L., et al. (2013). Twenty-first-century compatible CO₂ emissions and airborne fraction simulated by CMIP5 Earth System Models under four representative concentration pathways. *Journal of Climate*, 26(13), 4398–4413. https://doi.org/10.1175/JCLI-D-12-00554.1
- Kaduk, J. L., & Los, S.O. (2011). Predicting the time of green up in temperate and boreal biomes. Climatic Change, 107, 277–304. https://doi.org/10.1007/s10584-010-9951-1
- Kang, L., Han, X., Zhang, Z., & Sun, O. J. (2007). Grassland ecosystems in china: review of current knowledge and research advancement. Philosophical Transactions of the Royal Society, B: Biological Sciences, 362(1482), 997–1008. https://doi.org/10.1098/rstb.2007.2029
- Keenan, T., Baker, I., Barr, A., Ciais, P., Davis, K., Dietze, M., et al. (2012). Terrestrial biosphere model performance for inter-annual variability of land-atmosphere CO₂ exchange. *Global Change Biology*, 18(6), 1971–1987. https://doi.org/10.1111/j.1365-2486.2012. 02678.x
- Kim, Y., Moorcroft, P. R., Aleinov, I., Puma, M. J., & Kiang, N. Y. (2015). Variability of phenology and fluxes of water and carbon with observed and simulated soil moisture in the Ent Terrestrial Biosphere Model (Ent TBM version 1.0.1.0.0). Geoscientific Model Development, 8(12), 3837–3865. https://doi.org/10.5194/gmd-8-3837-2015
- Klein, T. H., & Hoch, G. (2015). Tree carbon allocation dynamics determined using a carbon mass balance approach. *The New Phytologist*, 205(1), 147–159. https://doi.org/10.1111/nph.12993
- Knapp, A., Briggs, J., Hartnett, D., & Collins, S. (1998). Grassland Dynamics: Long-Term Ecological Research in Tallgrass Prairie (p. 364). New York: Oxford University Press.
- Knapp, A. K., Hoover, D. L., Blair, J. M., Buis, G., Burkepile, D. E., Chamberlain, A., et al. (2012). A test of two mechanisms proposed to optimize grassland aboveground primary productivity in response to grazing. *Journal of Plant Ecology*, 5(4), 357–365. https://doi.org/10. 1093/jpe/rts020
- Koerner, S., Burkepile, D., Fynn, R., Burns, C., Eby, S., Govender, N., et al. (2014). Plant community response to loss of large herbivores differs between North American and South American savanna grasslands. *Ecology*, *95*(4), 808–816.
- Krinner, G., Viovy, N., Noblet-Ducoudré, N., Ogée, J., Friedlingsten, P., Ciais, P., et al. (2005). A dynamic global vegetation model for studies of the coupled atmosphere-biosphere system. *Global Biogeochemical Cycles*, 19, GB1015. https://doi.org/10.1029/2003GB002199
- Larcher, W. (2002). Plants under stress. In W. Larcher (Ed.), *Physiological plant ecology: Ecophysiology and stress physiology of functional groups* (pp. 345–450). Berlin, Heidelberg: Springer-Verlag.
- Lawrence, D. M., Oleson, K. W., Flanner, M. G., Fletcher, C. G., Lawrence, P. J., Levis, S., et al. (2012). The CCSM4 land simulation, 1850-2005: Assessment of surface climate and new capabilities. *Journal of Climate*, 25(7), 2240–2260. https://doi.org/10.1175/ JCLI-D-11-00103.1
- Lawrence, D. M., Oleson, K. W., Flanner, M. G., Thornton, P. E., Swenson, S. C., Lawrence, P. J., et al. (2011). Parameterization improvements and functional and structural advances in version 4 of the Community Land Model. *Journal of Advances in Modeling Earth Systems*, 3, M03001. https://doi.org/10.1029/2011MS00045
- Liang, L., and K. He (2017), Call for content for special issue: Patterns, trends, and ecological applications of phenology, *Remote Sensing in Ecology and Conservation Blog.*
- Lokupitiya, E., Denning, A., Paustian, K., Baker, I., Schaefer, K., Verma, S., et al. (2009). Incorporation of crop phenology in Simple Biosphere Model (SiBcrop) to improve land-atmosphere carbon exchanges from croplands. *Biogeosciences*, *6*, 969–986.
- McCormack, M., Adams, T., Smithwick, E., & Eissenstat, D. (2014). Variability in root production, phenology, and turnover rate among 12 temperate tree species. *Ecology*, 95(8), 2224–2235.
- Medvigy, D., Wofsy, S. C., Munger, J. W., Hollinger, D. Y., & Moorcroft, P. R. (2009). Mechanistic scaling of ecosystem function and dynamics in space and time: Ecosystem Demography model version 2. Journal of Geophysical Research, 114, G01002. https://doi.org/10.1029/ 2008JG000812



- Melaas, E. K., Friedl, M. A., & Richardson, A. D. (2016). Multiscale modeling of spring phenology across deciduous forests in the eastern United States. *Global Change Biology*, 22(2), 792–805. https://doi.org/10.1111/gcb.13122
- Migliavacca, M., Sonnentag, O., Keenan, T. F., Cescatti, A., O'Keefe, J., & Richardson, A. D. (2012). On the uncertainty of phenological responses to climate change, and implications for a terrestrial biosphere model. *Biogeosciences*, 9(6), 2063–2083. https://doi.org/10.5194/bg-9-2063-2012
- Mikan, C., Schimel, J., & Doyle, A. (2002). Temperature controls of microbial respiration in arctic tundra soils above and below freezing. Soil Biological Biochemistry, 34(11), 1785–1795. https://doi.org/10.1016/S0038-0717(02)00168-2
- Molchanov, A. (2009). Effect of moisture availability on photosynthetic productivity and autotrophic respiration of an oak stand. *Russian Journal of Plant Physiology*, 56(6), 769–779. https://doi.org/10.1134/S1021443709060065
- Morellato, L., Alberton, B., Alvarado, S., Borges, B., Buisson, E., Camargo, M., et al. (2016). Linking plant phenology to conservation biology. Biological Conservation, 195, 60–72. https://doi.org/10.1016/j.biocon.2015.12.033
- Negrón-Juárez, R., Koven, C., Riley, W., Knox, R. C., & Chambers, J. Q. (2015). Observed allocations of productivity and biomass, and turnover times in tropical forests are not accurately represented in CMIP5 Earth system models. *Environmental Research Letters*, 10, 064017. https://doi.org/10.1088/1748-9326/10/6/064017
- ORNL DAAC (2008a), MODIS Collection 5 Land Products Global Subsetting and Visualization Tool, ORNL DAAC, Oak Ridge, Tennessee, USA. Accessed August 2016. Subsets obtained for MOD15A2 product at site 39.08N, 96.56W, time period: 2000-2-18 to 2014-12-31, and subset size: 0.25x0.25 km, https://doi.org/10.334/ORNLDAAC/1241, 10.334/ORNLDAAC/1241.
- ORNL DAAC (2008b), MODIS Collection 5 Land Products Global Subsetting and Visualization Tool, ORNL DAAC, Oak Ridge, Tennessee, USA. Accessed August 2016. Subsets obtained for MYD15A2 product at site 39.08N, 96.56W, time period: 2000-2-18 to 2014-12-31, and subset size: 0.25x0.25 km, https://doi.org/10.334/ORNLDAAC/1241, 10.334/ORNLDAAC/1241.
- Pau, S., Wolkovich, E., Cook, B., Davies, J., Kraft, N., Bolmgren, K., et al. (2011). Predicting phenology by integrating ecology, evolution and climate science. *Global Change Biology*, 17(12), 3633–3643. https://doi.org/10.1111/j.1365-2486.2011.02515.x
- Picard, G., Quegan, S., Delbart, N., Lomas, M., Le Toan, T., & Woodward, F. (2005). Bud-burst modelling in Siberia and its impact on quantifying the carbon budget. *Global Change Biology*, 11(12), 2164–2176. https://doi.org/10.1111/j.1365-2486.2005.01055.x
- Reed, B., White, M., & Brown, J. (2003). Remote sensing phenology. In M. Schwartz (Ed.), *Phenology: An Integrative Environmental Science* (pp. 365–378). Germany: Dordrecht.
- Restrepo-Coupe, N., da Rocha, H. R., Hutyra, L. R., da Araujo, A. C., Borma, L. S., Christoffersen, B., et al. (2013). What drives the seasonality of photosynthesis across the Amazon basin? A cross-site analysis of eddy flux tower measurements from the Brasil flux network. *Agricultural and Forest Meteorology*, 182-183, 128–144. https://doi.org/10.1016/j.agrformet.2013.04.031
- Richardson, A. D., Anderson, R. S., Arain, M. A., Barr, A. G., Bohrer, G., Chen, G., et al. (2012). Terrestrial biosphere models need better representation of vegetation phenology: Results from the North American Carbon Program site synthesis. *Global Change Biology*, 18(2), 566–584. https://doi.org/10.1111/j.1365-2486.2011.02562.x
- Richardson, A. D., Keenan, T., Migliavacca, M., Ryu, Y., Sonnentag, O. T., & Toomey, M. (2013). Climate change, phenology, and phenological control of vegetation feedbacks to the climate system. Agricultural and Forest Meteorology, 169, 156–173. https://doi.org/10j.1016/j. agrformet.2012.09.012
- Richardson, A. D., Weltzin, J., & Morisette, J. (2017). Integrating multiscale seasonal data for resource management. *Eos*, 98. https://doi. org/10.1029/2017EO065709
- Riley, W. J., Gaudinski, J. B., Torn, M. S., Joslin, J. D., & Hanson, P. J. (2009). Fine-root mortality rates in a temperate forest: Estimates using radiocarbon data and numerical modeling. *New Phytologist*, 184(2), 387–398. https://doi.org/10.1111/j.1469-8137.2009.02980.x
- Rishmawi, K., Prince, S., & Xue, Y. (2016). Vegetation responses to climate variability in the northern arid to sub-humid zones of Sub-Saharan Africa. *Remote Sensing*, 8(11), 910. https://doi.org/10.3390/rs8110910
- Sala, O. E., Parton, W. J., Joyce, L. A., & Lauenroth, W. K. (1988). Primary production of the central grassland region of the United States. Ecology, 69(1), 40–45. https://doi.org/10.2307/1943158
- Saleska, S., Wu, J., Guan, K., Araujo, A., Huerte, A., Nobre, A., & Restrepo-Coupe, N. (2014). Dry-season greening of Amazon forests. *Nature*, 531. https://doi.org/10.1038/nature16457
- Schaefer, K., Collatz, G. J., Tans, P., Denning, A. S., Baker, I., Berry, J., et al. (2008). Combined simple biosphere/Carnegie-Ames-Stanford Approach terrestrial carbon cycle model. *Journal of Geophysical Research*, 113, G03034. https://doi.org/10.1029/2007JG000603
- Schwartz, M. (Ed) (2013). Phenology: An integrative environmental science. New York, USA: Springer. https://doi.org/10.1007/ 978-94-007-6925-0
- Sellers, P. (1987). Canopy reflectance, photosynthesis, and transpiration. II. The role of biophysics in the linearity of their interdependence. *Remote Sensing of Environment*, *21*, 143–183.
- Sellers, P., Berry, J., Collatz, G., Field, C., & Hall, F. (1992). Canopy reflectance, photosynthesis, and transpiration. III. A reanalysis using improved leaf models and a new canopy integration scheme. *Remote Sensing of Environment*, 42, 187–216.
- Sellers, P., Los, S., Tucker, C., Justice, C., Dazlich, D., Collatz, G., & Randall, D. (1996). A revised land surface parameterization (SiB2) for atmospheric GCMs Part II: The generation of global fields of terrestrial biophysical parameters from satellite data. *Journal of Climate*, *9*, 706–737.
- Sellers, P., Mintz, Y., Sud, Y., & Dalcher, A. (1986). A Simple Biosphere Model (SiB) for use within general circulation models. *Journal of Atmospheric Science*, 43(6), 505–531.
- Sellers, P., Randall, D., Collatz, G., Berry, J., Field, C., Dazlich, D., et al. (1996). A revised land surface parameterization (SiB2) for atmospheric GCMs. *Journal of Climate*, 9, 676–705.
- Smith, D., A. Denning, M. Smith, C. O'Dell, and C. Kummerow (2018), Using remotely sensed fluorescence and soil moisture to better understand the seasonal cycle of tropical grasslands, Master's thesis, Colorado State University, Fort Collins, Colorado, USA.
- Smithwick, E., Lucash, M., McCormack, M., & Sivandran, G. (2014). Improving the representation of roots in terrestrial models. Environmental Science and Management Faculty Publications and Presentations, 108.
- Stöckli, R., Rutishauser, T., Dragoni, D., O'Keefe, J., Thornton, P., Jolly, M., et al. (2008). Remote sensing data assimilation for a prognostic phenology model. *Journal of Geophysical Research*, 113, G04021. https://doi.org/10.1029/2008JG000781
- Stöckli, R., Rutishauser, T., Dragoni, D., O'Keefe, J., Thornton, P., Jolly, M., et al. (2011). A global reanalysis of vegetation phenology. Journal of Geophysical Research, 116, G03020. https://doi.org/10.1029/2010JG001545
- Suzuki, Y., Kihara-Doi, T., Kawazu, T., Miyake, C., & Makino, A. (2010). Differences in rubisco content and its synthesis in leaves at different positions in *eucalyptus globulus* seedlings. *Plant, Cell and Environment*, 33, 1314–1323. https://doi.org/10.1111/j.1365-3040.2010.02149.x



- Urrutia-Jalabert, R., Malhi, Y., & Lara, A. (2017). Soil respiration and mass balance estimation of fine root production in fitzroya cupressoides forests of southern Chile. *Ecosphere*, 8(4), e01640. https://doi.org/10.1002/ecs2.1640
- White, M., Thornton, P., & Running, S. (1997). A continental phenology model for monitoring vegetation responses to interannual climatic variability. *Global Biogeochemical Cycles*, 11(2), 217–234.
- Wilson, K., & Baldocchi, D. (2000). Estimating annual net ecosystem exchange of carbon over five years at a deciduous forest in the southern United States. Agricultural and Forest Meteorology, 100, 1–18.
- Wu, J., Albert, L., Lopes, A., Restrepo-Coupe, N., Hayek, M., Wiedermann, K., et al. (2016). Leaf development and demography explain photosynthetic seasonality in Amazon evergreen forests. *Science*, 351(6276), 13–25. https://doi.org/10.1126/science.aad5068
- Xia, J., Yuan, W., Wang, Y.-P., & Zhang, Q. (2017). Adaptive carbon allocation by plants enhances the terrestrial carbon sink. Scientific Reports, 7(3341), 1–16. https://doi.org/10.1038/s41598-017-03574-3
- Xu, X., Medvigy, D., Joseph Wright, S., Kitajima, K., Wu, J., Albert, L. P., et al. (2017). Variations of leaf longevity in tropical moist forests predicted by a trait-driven carbon optimality model. *Ecology Letters*, 20(9), 1097–1106. https://doi.org/10.1111/ele.12804
- Yan, L., Zhou, G., & Zhang, F. (2013). Effects of different grazing intensities on grassland production in China: A meta-analysis. *PLoS ONE*, 8(12). https://doi.org/10.1371/journal.pone.0081466
- Zhang, X., Friedl, M., Schaaf, C., Strahler, A., Hodges, J., Gao, F., et al. (2003). Monitoring vegetation phenology using MODIS. *Remote Sensing of Environment*, 84, 471–475. https://doi.org/10.1016/S0034-4257(02)00135-9
- Zhou, Y., Ju, W., Sun, X., Hu, Z., Han, S., Black, T. A., et al. (2014). Close relationship between spectral vegetation indices and V_{cmax} in deciduous and mixed forests. *Tellus Series B: Chemical and Physical Meteorology*, 66, 23279. https://doi.org/10.3402/tellusb.v66.23279